Author’s response to reviews

Title: Prevalence of behavioral disorders and Attention deficit Hyperactive disorder among school going children in Southwestern Uganda

Authors:

Apollo Kivumbi (apoku@live.com)
William Byansi (byansiw@wustl.edu)
Christopher Damulira (ddamchrist1@gmail.com)
James Mugisha (jmmugi77@hotmail.com)
Phionah Namatovu (namatovuphionah7@gmail.com)
Ozge Bahar (ozge.sensoybahr@wustl.edu)
Mary Mckay (mary.mckay@wustl.edu)
Kimberly Hoagwood (kimberly.hoagwood@nyumc.org)
Fred Ssewamala (fms1@wustl.edu)

Version: 4 Date: 13 Feb 2019

Author’s response to reviews:

Thank you for this comment- p-values and chi squares have been included in table 4 like other tables